Author's response to reviews

Title: Single Lung Mass in a Patient with a History of Lymphoma: a case report

Authors:

Ying Yuan (yuanying1999@zju.edu.cn)
Hong Shen (shenhong1983@126.com)
Hanguang Hu (huhanguang@zju.edu.cn)
Xiaoxian Ye (yexx1985@163.com)
Xian Zhong (xzhong9@gmail.com)

Version: 3 Date: 20 March 2013

Author's response to reviews:

Please include the ethnicity of the patient in the abstract and in the main case presentation section of the manuscript. (Eg. Caucasian)

- Please replace dates found in the case presentation to the amount of time this occurred before the presentation of the case report.

- Please remove any text from figure 1 which may reveal the patient’s identity (e.g. patient’s number, patient’s name, dates, etc)

These have been corrected.